Author's response to reviews

Title: Use of drug-eluting balloon coronary intervention prior to living donor kidney transplantation.

Authors:

Tobias Kammerer (tobias.kammerer@med.uni-muenchen.de)
Andres Beiras-Fernandez (andres.Beiras@kgu.de)
Markus Rehm (markus.rehm@med.uni-muenchen.de)
Manfred Stangl (manfred.stangl@med.uni-muenchen.de)
Markus Guba (markus.guba@med.uni-muenchen.de)
Christian Kupatt (christian.kupatt@med.uni-muenchen.de)
Florian Weis (florian.weis@klinikum-ffb.de)

Version: 6  Date: 15 August 2014

Author’s response to reviews:

In the abstract there was confusion with the last sentence of the background. It was revised to: "Among them, coronary artery disease is a growing problem. An increasing number of patients, undergoing living donation, receive antiplatelet therapies due to coronary disease."

The patients’ consent was clarified by the following sentence: "Written informed consent was obtained from the patient for publication of this Case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal."